Author’s response to reviews

Title: Process of care in outpatient Integrative healthcare facilities: a systematic review of clinical trials.

Authors:

Suzanne J Grant (s.grant@uws.edu.au)
Jane Frawley (Jane.E.Frawley@student.uts.edu.au)
Alan Bensoussan (a.bensoussan@uws.edu.au)

Version: 5 Date: 14 July 2015

Author’s response to reviews: see over

Suzanne J. Grant¹, Jane Frawley², Alan Bensoussan¹

¹National Institute of Complementary Medicine, University of Western Sydney, Locked Bag 1797, Penrith, NSW 2751 Australia

²Australian Research Centre in Complementary and Integrative Medicine, University of Technology Sydney, Australia

§Corresponding author

Email addresses: SJG: s.grant@uws.edu.au

AB: a.bensoussan@uws.edu.au

JF: jane.e.frawley@student.uts.edu.au